Author’s response to reviews

Title: Setting strategy for system change: Using concept mapping to prioritise national action for chronic disease prevention.

Authors:

Sonia Wutzke (sonia.wutzke@saxinstitute.org.au)
Nick Roberts (nick.roberts@saxinstitute.org.au)
Cameron Willis (cameron.willis@sydney.edu.au)
Allan Best (allan.best@in-source.ca)
Andrew Wilson (a.wilson@sydney.edu.au)
William Trochim (wmt1@cornell.edu)

Version: 2 Date: 24 Jun 2017

Author’s response to reviews:

Please see attached letter.